Author's response to reviews

Title: A randomised trial of a brief cognitive-behavioural, manualised, self-management programme for hypertension delivered in a cardiac patient club in Shanghai

Authors:

Feng Xue (fx101@york.ac.uk)
Wen Yao (yw612423@163.com)
Robert J Lewin (RJPL1@york.ac.uk)

Version: 2 Date: 10 August 2007

Author's response to reviews:

Dear Editor,
In the revised manuscript Table 2 stays on one page.
Regards,
Feng Xue